Author's response to reviews

Title: A Systematic Comparison and Evaluation of High Density Exon Arrays and RNA-seq Technology used to Unravel the Peripheral Blood Transcriptome of Sickle Cell Disease.

Authors:

Nalini Raghavachari (nraghavachari@nhlbi.nih.gov)
Jennifer Barb (barbij@mail.nih.gov)
Yanqin Yang (Yanqin.yang@mail.nih.gov)
Poching Liu (pcliu@nhlbi.nih.gov)
Kimberly Woodhouse (kwoodhous@nhlbi.nih.gov)
Daniel Levy (levyd@nhlbi.nih.gov)
Chris O'donnell (odonnellc@nhlbi.nih.gov)
Peter J Munson (munson@mail.nih.gov)
Gregory Kato (gkato@nhlbi.nih.gov)

Version: 4 Date: 27 January 2012

Author's response to reviews: see over
January 27, 2012

Ms. Elaine Patricia Cruz  
Journal Editorial Office  
BioMed Central  
236 Gray's Inn Road  
London WC1X 8HB  
United Kingdom

Dear Ms. Cruz,

I am pleased to submit herewith the revised manuscript entitled “A Systematic Comparison and Evaluation of High Density Exon Arrays and RNA-seq technology in Unraveling the Peripheral Blood Transcriptome of Sickle Cell Disease” addressing the reviewer’s comments for publication in BMC Medical Genomics. I have also uploaded a file entitled Response to reviewer’s comments detailing the corrections that we have made in this article based on their suggestions.

We believe this manuscript will be of great interest to the readership of BMC Medical Genomics and will have a broad impact for clinical researchers and help them decide on the choice of tools in the form of arrays or RNA-seq for clinical transcriptomic studies.

Looking forward for a favorable response from you.

Thank You.

Sincerely

Nalini Raghavachari, Ph. D